Author’s response to reviews

Title: The effects of essential oil mouthrinses with or without alcohol on plaque and gingivitis: a randomized controlled clinical study

Authors:

Michael Lynch (mlynch23@its.jnj.com)
Sheila Cortelli (cavalcacortelli@uol.com.br)
James McGuire (JMcGuire@its.jnj.com)
Jane Zhang (zhang07920@yahoo.com)
Danette Ricci-Nittel (dnittel@its.jnj.com)
Carolyn Mordas (cmordas@its.jnj.com)
Davi Aquino (daviaquino@uol.com.br)
Jose Cortelli (jrcortelli@uol.com.br)

Version: 3 Date: 20 Nov 2017

Author’s response to reviews:

Dear Editor,

Thank you for reviewing our manuscript. Below is a point-by-point response to your comments:

1. At the end of the abstract, within the trial registration details please ensure to provide a date of the trial registration.

Trial Registration: The trial was registered on clinicaltrials.gov on November 30, 2016. The registration number is NCT02980497.

2. Please provide authors' initials in the 'Authors' contributions' section in the Declarations.

Dr. Lynch (MCL), Dr. Sheila Cortelli (SCC), Ms. Ricci-Nittel (DRN), Dr. Aquino (DRA) and Dr. Jose Cortelli (JRC) contributed to the design and execution of the study. Mr. McGuire (JAM) contributed to the study design and statistical analysis. Dr. Mordas contributed (CJM) to the formulation and related discussion. Dr. Zhang (JZ) led the development of the manuscript and served as the corresponding author. All authors contributed to the development of the manuscript and approved the manuscript.
3. On uploading your revisions, please remove any tracked changes or highlighting and include only a single clean copy.

No tracked change or highlighting were included.

We truly appreciate your time and effort reviewing our manuscript and look forward to hearing from you again. Thanks.

Jane Zhang, Ph. D